Author's response to reviews

Title: Lingual juvenile xanthogranuloma in an adult patient. A rare case report.

Authors:

alessandro villa (villales@hotmail.it)
umberto mariani (marianiumberto@libero.it)
francesco villa (fvilla@ospedaliriuniti.bergamo.it)

Version: 3 Date: 19 July 2010

Author's response to reviews: see over
Dear Editor,

We are thankful for the reviewers’ thoughtful comments. We have considered them carefully and revised the manuscript accordingly. Our point-by-point responses are given below. Please find enclosed the revised manuscript with track changes. Please, let me know if you have further questions.

Thank you again for your support.

Kind regards,

Alessandro Villa

Alessandro Villa, D.D.S.
Department of Dentistry.
Azienda Ospedaliera Ospedali Riuniti di Bergamo.
Largo Barozzi 1.
24128, Bergamo, Italy
Phone: (+39) 393-6900258
villales@hotmail.it
Note: All the changes are “highlighted in yellow”.

Reviewer #1

This reviewer asked no more corrections.

Reviewer #2

All linguistic corrections have been made as requested by the referee.